Author's response to reviews

Title: Spectral analysis of HIV seropositivity among migrant workers entering Kuwait

Authors:

Saeed Akhtar (saeed.akhtar@hsc.edu.kw)
Hameed GH. H. Mohammad (hameed@kma.org.kw)

Version: 4 Date: 13 March 2008

Author's response to reviews: see over
March 13, 2008

Revised MS: 3173322691602549

Spectral analysis of HIV seropositivity among migrant workers entering Kuwait
Saeed Akhtar and Hameed GH. H. Mohammad

Dear Colleagues

Please find uploaded herewith above referred manuscript, which has been revised as follow:

i. Page 3: Word ‘Introduction’ has been replaced with word ‘Background’
ii. Page 8: Word ‘Figure1’ has been replaced by word ‘Figure 1”
iii. Page 12: A heading titled “Conclusion” has been added.
iv. Figure cropping: Figure 1 and Figure 2 have been further cropped.

Thank you for your time and considerations.

With best regards,

-sd-
Saeed Akhtar, PhD
Associate Professor, Epidemiology
Department of Community Medicine & Behavioral Sciences,
Faculty of Medicine, Kuwait University
PO Box 24923, Safat 13110, Kuwait

Phone: (965) 4986542
Fax: (965) 5338948
Email: saeed.akhtar@hsc.edu.kw